Author's response to reviews

Title: Multiple mechanisms underlying acquired resistance to taxanes in selected docetaxel-resistant MCF-7 breast cancer cells

Authors:

Harris Wang (hcw@ualberta.ca)
The Vo (tmv@ualberta.ca)
Ali Hajar (ahajar@telus.net)
Sara Li (saraazjli@gmail.com)
Xinmei Chen (xinmei.chen@ualberta.ca)
Amadeo M Parissenti (aparissenti@amric.ca)
David N Brindley (david.brindley@ualberta.ca)
Zhixiang Wang (zhixiang.wang@ualberta.ca)

Version: 2
Date: 14 January 2014

Author's response to reviews: see over
January 14, 2014

Dr. Dafne Solera  
Editor, BMC Cancer

Dear Dr. Solera:

Thank you for your letter regarding our manuscript entitled “Multiple mechanisms underlying acquired resistance to taxanes in selected docetaxel-resistant MCF-7 breast cancer cells.” We are grateful that the manuscript is accepted pending minor revisions.

We have revised our manuscript by including the following sections indicated in your letter: Conflict of interest declaration, Author’s contribution, and Acknowledgements.

Thank you.

Yours sincerely,

Zhixiang Wang  
Professor